Author's response to reviews

Title: A prospective cohort study of the association between drinking water arsenic exposure and self-reported maternal health symptoms during pregnancy in Bangladesh

Authors:

Molly L Kile (molly.kile@oregonstate.edu)
Ema Rodrigues (emarod@hsph.harvard.edu)
Maitreyi Mazumdar (maitreyi.mazumdar@childrens.harvard.edu)
Christine Dobson (cdobson@hsph.harvard.edu)
Nancy Diao (ndiao@hsph.harvard.edu)
Mostofa Golam (mostofa07@gmail.com)
Quazi Quamruzzaman (qqz@bangla.net)
Mahmudar Rahman (dch@bangla.net)
David C Christiani (dchris@hsph.harvard.edu)

Version: 2 Date: 11 November 2013

Author's response to reviews:

The editor requested the following formatting changes prior to initiating the peer review process.
1. Include a title page as page 1 that includes email addresses of all authors
2. Include an abstract on page 2

We have made these requested changes.